Author's response to reviews


Authors:

    Onno de Zwart (dezwarto@ggd.rotterdam.nl)
    Irene K. Veldhuijzen (veldhuijzeni@ggd.rotterdam.nl)
    Jan Hendrik Richardus (richardush@ggd.rotterdam.nl)
    Johannes Brug (j.brug@vumc.nl)

Version: 5  Date: 26 April 2010

Author's response to reviews:

Rotterdam, April 26, 2010

Dear editor,

Thank you very much for the acceptance of our manuscript ‘Monitoring of risk perceptions and correlates of precautionary behaviour related to human avian influenza during 2006 – 2007 in the Netherlands: results of seven consecutive surveys.’ We have made the last required editorial requests.

- Affiliations. The necessary information has been added.
- The numbering of the figures has been altered. The first figure cited in the text is now figure 1.
- The table which was in landscape has been changed in to a portrait format.
- The title of the introduction has been changed into background.
- The heading references has been included.
- The vertical lines of the tables have been removed.

We feel we have made all the necessary editorial changes. If there are any further questions please do not hesitate to get in touch

Yours sincerely,

Best wishes,

Onno de Zwart